Author's response to reviews

Title: Gene Expression Analysis Reveals Impaired Mitochondriogenesis and Adipogenesis in Adipose Tissue from a Patient with Acquired Partial Lipodystrophy (Barraquer-Simons Syndrome): a case report

Authors:

Jordi P Guallar (jordi.gualler@gmail.com)
Ricardo Rojas-Garcia (rrojas@santpau.es)
Elena Garcia-Arumi (egarumi@ir.vhebron.net)
Joan C Domingo (jcdomingo@ub.edu)
Eduardo Gallardo (egallardo@santpau.es)
Antoni L Andreu (aandreu@vhebron.net)
Pere Domingo (pdomingo@santpau.es)
Isabel Illa (iilla@santpau.es)
Marta Giralt (mgiralt@ub.edu)
Francesc Villarroya (fvillarroya@ub.edu)

Version: 2 Date: 12 March 2008

Author's response to reviews:

This is a first revision of formal presentation issues requested by your editorial office in order to fulfil the journal style.

Revisions are:

We have moved details of funding to a section 'Acknowledgements' before the Reference list.

Abstract is now shown on page 2 of the manuscript.

The country of origin and ethnic origin of the patient is shown (see page 5).

A list of abbreviations used in the manuscript and their meanings is shown in the beginning of the consent section.

The contribution of Eduardo Gallardo is shown now (see second line in Author’s contribution, page 10). It was not described in the first version because of a clerical mistake (confusion between initials EG for Eduardo Gallardo and EGA for Elena Garcia-Arumi).

I hope this formal issues are ok now,

Yours sincerely,
Francesc Villarroya

In accordance with your request we have:

Moved details of funding to a section 'Acknowledgements' before the Reference list.

Abstract is now shown on page 2 of the manuscript.

The country of origin and ethnic origin of the patient is shown (see page 5).

A list of abbreviations used in the manuscript and their meanings is shown (see page 3)

The contribution of Eduardo Gallardo is shown now.